**Author’s response to reviews**

**Title:** Defining the short-term disease recurrence after Loop Electrosurgical Excision Procedure (LEEP)

**Authors:**

Nicholas Papalia (nicholas.papalia@albertahealthservices.ca)

Amanda Rohla (amanda.rohla@ahs.ca)

Selpee Tang (Selpee.Tang@albertahealthservices.ca)

Jill Nation (jill.nation@ahs.ca)

Gregg Nelson (Gregory.Nelson@albertahealthservices.ca)

**Version:** 2  **Date:** 31 Jan 2020

**Author’s response to reviews:**

All editors revisions have been included (Author contributions, author initials, consent to participate, acknowledgements, declaration, formatting, and removal of supplementary files).

Reviewer #1 - all 4 revisions addressed

Added consent for publishing. Added competing interests section.